Author’s response to reviews

Title: The impact of Cytochrome 450 and Paraoxonase polymorphisms on clopidogrel resistance and major adverse cardiac events in Coronary Heart Disease patients after Percutaneous Coronary Intervention

Authors:

zhaowei zhang (quhaizhijia@163.com)

minxiao chen (15358693935@163.com)

Long Zhang (392814297@qq.com)

Qiang Zhao (314987232@qq.com)

Version: 2 Date: 14 Dec 2019

Author’s response to reviews:

Dear Editor:

We are extremely thankful for the invaluable comments we have received from you. We have therefore revised the manuscript. Firstly, we rephrase the overlapped text in the Methods (p4, p5), Results (p6-7) as you have mentioned. We hope it will meet the requirement. Secondly, We represent authors' names using their full initials in the Authors’ Contributions section. Lastly, We revised manuscript as a clean copy without any tracked changes, coloured or highlighted text.

Sincerely,

Zhaowei Zhang